%~MoulinÃ© par MaN_auto v.0.33.8 (9c3f6e85) 2025-05-26 17:24:51
\documentclass[CRMATH,Unicode,XML,biblatex]{cedram}

\TopicFR{ProbabilitÃ©s}
\TopicEN{Probability theory}

\addbibresource{CRMATH_Li_20240726.bib}

\usepackage{mathtools}

\newcommand*\RR{\mathbb{R}}
\newcommand*\EE{\mathbb{E}}
\newcommand*\be{{\beta}}
\newcommand*\de{{\delta}}
\newcommand*\la{{\lambda}}
\newcommand*\si{{\sigma}}
\newcommand*\De{{\Delta}}
\newcommand*\al{{\alpha}}

\newcommand{\KK}{\mathcal K}
\newcommand{\HH}{\mathcal H}
\newcommand{\DD}{\mathbb D}
\newcommand{\QQ}{\mathcal Q_0}

\newcommand*{\twave}{{\smash{{\underset{\raisebox{1ex}{$\sim$}}{t}}}\vphantom{\underline{t}}}}
\newcommand*{\twavesub}{{\smash{{\underset{\raisebox{1ex}{$\scriptstyle\sim$}}{t}}}\vphantom{\underline{t}}}}

\newcommand*{\dd}{\mathrm{d}}

\DeclareMathOperator{\Trace}{Trace}
\DeclareMathOperator{\support}{support}
\DeclareMathOperator{\supp}{supp}

%%%---------------------------------------------------------------------------

%%% BAR & OVERLINE

%%% Better widebar macro
\makeatletter
\let\save@mathaccent\mathaccent
\newcommand*\if@single[3]{%
  \setbox0\hbox{${\mathaccent"0362{#1}}^H$}%
  \setbox2\hbox{${\mathaccent"0362{\kern0pt#1}}^H$}%
  \ifdim\ht0=\ht2 #3\else #2\fi
  }
%The bar will be moved to the right by a half of \macc@kerna, which is computed by amsmath:
\newcommand*\rel@kern[1]{\kern#1\dimexpr\macc@kerna}
%If there's a superscript following the bar, then no negative kern may follow the bar;
%an additional {} makes sure that the superscript is high enough in this case:
\newcommand*\widebar[1]{\@ifnextchar^{{\wide@bar{#1}{0}}}{\wide@bar{#1}{1}}}
%Use a separate algorithm for single symbols:
\newcommand*\wide@bar[2]{\if@single{#1}{\wide@bar@{#1}{#2}{1}}{\wide@bar@{#1}{#2}{2}}}
\newcommand*\wide@bar@[3]{%
  \begingroup
  \def\mathaccent##1##2{%
%Enable nesting of accents:
    \let\mathaccent\save@mathaccent
%If there's more than a single symbol, use the first character instead (see below):
    \if#32 \let\macc@nucleus\first@char \fi
%Determine the italic correction:
    \setbox\z@\hbox{$\macc@style{\macc@nucleus}_{}$}%
    \setbox\tw@\hbox{$\macc@style{\macc@nucleus}{}_{}$}%
    \dimen@\wd\tw@
    \advance\dimen@-\wd\z@
%Now \dimen@ is the italic correction of the symbol.
    \divide\dimen@ 3
    \@tempdima\wd\tw@
    \advance\@tempdima-\scriptspace
%Now \@tempdima is the width of the symbol.
    \divide\@tempdima 10
    \advance\dimen@-\@tempdima
%Now \dimen@ = (italic correction / 3) - (Breite / 10)
    \ifdim\dimen@>\z@ \dimen@0pt\fi
%The bar will be shortened in the case \dimen@<0 !
    \rel@kern{0.6}\kern-\dimen@
    \if#31
      \overline{\rel@kern{-0.6}\kern\dimen@\macc@nucleus\rel@kern{0.4}\kern\dimen@}%
      \advance\dimen@0.4\dimexpr\macc@kerna
%Place the combined final kern (-\dimen@) if it is >0 or if a superscript follows:
      \let\final@kern#2%
      \ifdim\dimen@<\z@ \let\final@kern1\fi
      \if\final@kern1 \kern-\dimen@\fi
    \else
      \overline{\rel@kern{-0.6}\kern\dimen@#1}%
    \fi
  }%
  \macc@depth\@ne
  \let\math@bgroup\@empty \let\math@egroup\macc@set@skewchar
  \mathsurround\z@ \frozen@everymath{\mathgroup\macc@group\relax}%
  \macc@set@skewchar\relax
  \let\mathaccentV\macc@nested@a
%The following initialises \macc@kerna and calls \mathaccent:
  \if#31
    \macc@nested@a\relax111{#1}%
  \else
%If the argument consists of more than one symbol, and if the first token is
%a letter, use that letter for the computations:
    \def\gobble@till@marker##1\endmarker{}%
    \futurelet\first@char\gobble@till@marker#1\endmarker
    \ifcat\noexpand\first@char A\else
      \def\first@char{}%
    \fi
    \macc@nested@a\relax111{\first@char}%
  \fi
  \endgroup
}
\makeatother

\let\oldbar\bar
\renewcommand*{\bar}[1]{{\mathchoice{\widebar{#1}}{\widebar{#1}}{\widebar{#1}}{\oldbar{#1}}}}

% /!\ Il y a parfois des soucis avec cette macro lorsqu'il y a des caractÃ¨res en minuscule en subscript (et dans les captions). Dans ce cas, faire avec \overline et des espaces manuels.

%%% Double bar
% \overline{\overline{\!a}}

%\newcommand\bx{\,\overline{\!x}}
%\newcommand\brho{\,\overline{\!\rho}}
%better widebar a apparemment des soucis avec les \subcaptions pour les CR :shrug:

%%%---------------------------------------------------------------------------

%%% DELIMITERS

%\usepackage{mathtools}
\DeclarePairedDelimiter{\parens}{\lparen}{\rparen}
\DeclarePairedDelimiter{\braces}{\{}{\}}
\DeclarePairedDelimiter{\abs}{\lvert}{\rvert}
\DeclarePairedDelimiter{\norm}{\lVert}{\rVert}
\DeclarePairedDelimiter{\bracks}{[}{]}
\DeclarePairedDelimiter{\bbracks}{\llbracket}{\rrbracket}
\DeclarePairedDelimiter{\angles}{\langle}{\rangle}
\DeclarePairedDelimiter{\aangles}{\langle\!\langle}{\rangle\!\rangle}
\DeclarePairedDelimiter{\pparens}{(\!(}{)\!)}

%%%---------------------------------------------------------------------------

\graphicspath{{./figures/}}

\newcommand*{\mk}{\mkern -1mu}
\newcommand*{\Mk}{\mkern -2mu}
\newcommand*{\mK}{\mkern 1mu}
\newcommand*{\MK}{\mkern 2mu}

\hypersetup{urlcolor=purple, linkcolor=blue, citecolor=red}

\newcommand*{\relabel}{\renewcommand{\labelenumi}{(\theenumi)}}
\newcommand*{\romanenumi}{\renewcommand*{\theenumi}{\roman{enumi}}\relabel}
\newcommand*{\Romanenumi}{\renewcommand*{\theenumi}{\Roman{enumi}}\relabel}
\newcommand*{\alphenumi}{\renewcommand*{\theenumi}{\alph{enumi}}\relabel}
\newcommand*{\Alphenumi}{\renewcommand*{\theenumi}{\Alph{enumi}}\relabel}

\newcommand*{\alphenumii}{\renewcommand*{\theenumii}{\alph{enumii}}\renewcommand{\labelenumii}{(\theenumii)}}

\let\oldtilde\tilde
\renewcommand*{\tilde}[1]{\mathchoice{\widetilde{#1}}{\widetilde{#1}}{\oldtilde{#1}}{\oldtilde{#1}}}
\let\oldhat\hat
\renewcommand*{\hat}[1]{\mathchoice{\widehat{#1}}{\widehat{#1}}{\oldhat{#1}}{\oldhat{#1}}}

%%%---------------------------------------------------------------------------

%%% Changer la largeur des adresses

% Jouer avec \hsize :
\makeatletter
\def\@setafterauthor{%
  \vglue3mm%
%  \hspace*{0pt}%
\begingroup\hsize=12.5cm\advance\hsize\abstractmarginL\raggedright
\noindent
%\hspace*{\abstractmarginL}\begin{minipage}[t]{10cm}
   \leftskip\abstractmarginL
  \normalfont\Small
  \@afterauthor\par
\endgroup
\vskip2pt plus 3pt minus 1pt
}
\makeatother

%%%---------------------------------------------------------------------------

\title {Wong--Zakai approximations and support theorems for SDEs under Lyapunov conditions}
\alttitle {Approximations de Wong--Zakai et thÃ©orÃ¨mes de support pour les SDEs sous conditions de Lyapunov}

\author{\firstname{Qi} \lastname{Li}}
\email[Q.~Li]{vivien777@mail.ustc.edu.cn}
\address{School of Mathematical Sciences, University of Science and Technology of China, Hefei, Anhui 230026, China}

\author{\firstname{Jianliang} \lastname{Zhai}}
\email[J.~Zhai]{zhaijl@ustc.edu.cn}
\address[1]{School of Mathematical Sciences, University of Science and Technology of China, Hefei, Anhui 230026, China}

\author{\firstname{Tusheng} \lastname{Zhang}}
\email[T.~Zhang]{tusheng.zhang@manchester.ac.uk}
\address{Department of Mathematics, University of Manchester, Oxford Road, Manchester, M13 9PL, UK}

\keywords{\kwd{Wong--Zakai approximation} \kwd{support theorem} \kwd{local Lipschitz} \kwd{Lyapunov condition}}
\altkeywords{\kwd{Approximation de Wong--Zakai} \kwd{thÃ©orÃ¨me de support} \kwd{fonctions localement lipschitziennes} \kwd{condition de Lyapunov}}

\thanks{This work is partially supported by National Key R\& D program of China (No.~2022 YFA1006001)), National Natural Science Foundation of China (Nos.~12131019, 12371151, 11721101). Jianliang Zhai's research is also supported by the School Start-up Fund~(USTC) KY0010000036 and the Fundamental Research Funds for the Central Universities (No.~WK3470000016)}
\CDRGrant[National Key R\& D program of China]{2022 YFA1006001}
\CDRGrant[National Natural Science Foundation of China]{12131019}
\CDRGrant[National Natural Science Foundation of China]{12371151}
\CDRGrant[National Natural Science Foundation of China]{11721101}
\CDRGrant[School Start-up Fund~(USTC)]{KY0010000036}
\CDRGrant[Fundamental Research Funds for the Central Universities]{WK3470000016}

\begin{abstract}
In this paper, we establish the Stroock--Varadhan type support theorems for stochastic differential equations (SDEs) under Lyapunov conditions, which improve the existing results in the literature where the coefficients of the SDEs are required to be globally Lipschitz. Our conditions are mild to include many important models, e.g.\ Threshold Ornstein--Uhlenbeck process, stochastic SIR model, stochastic Lotka--Volterra systems, stochastic Duffing--van der Pol oscillator model, which have polynomial coefficients. To obtain the support theorem, a localizing procedure plays an important role.
\end{abstract}

\begin{altabstract}
Dans cet article, nous Ã©tablissons les thÃ©orÃ¨mes de support de type Stroock--Varadhan pour des Ã©quations diffÃ©rentielles stochastiques (EDS) sous conditions de Lyapunov, qui amÃ©liorent les rÃ©sultats existants dans la littÃ©rature oÃ¹ les coefficients des EDS doivent Ãªtre globalement Lipschitz. Nos conditions sont faibles et permettent d'inclure de nombreux modÃ¨les importants, par exemple le processus de Ornstein--Uhlenbeck avec seuillage, le modÃ¨le stochastique SIR, les systÃ¨mes stochastiques de Lotka--Volterra, le modÃ¨le stochastique de l'oscillateur de Duffing--van der Pol, qui ont des coefficients polynomiaux. Pour obtenir le thÃ©orÃ¨me de support, une procÃ©dure de localisation joue un rÃ´le important.
\end{altabstract}

\begin{document}

\maketitle

\section{Introduction}

Let $W$ denote a $d$-dimensional standard Wiener process on a complete filtered probability space $\parens[\big]{\Omega,\mathcal{F},(\mathcal{F}_t)_{t\geq0},\mathbb{P}}$, where $(\mathcal{F}_t)_{t\geq0}$ is the normal filtration generated by $W$. Denote by $\abs{\,\cdot\,}$, $\norm{\,\cdot\,}$ and $\langle\,\cdot\,,\cdot\,\rangle$ the $\mathbb{R}^m$-norm, $\mathbb{R}^m \otimes \mathbb{R}^d$-norm and inner product of $\mathbb{R}^m$, respectively. Without loss of generality, we work on the finite time interval $[0, 1]$. We denote by $\mathcal{C}\parens[\big]{[0,1];\mathbb{R}^m}$ the space of continuous functions $f \colon [0,1] \to \mathbb{R}^m$ with the norm $\abs{f}_{\infty}\coloneqq\sup_{t\in[0,1]}\abs{f_t}$. Let $\mathcal{H}$ denote the Cameron--Martin space, i.e., $\mathcal{H}\coloneqq\braces[\big]{ h:\dot{h}\in L^2\parens[\big]{[0,1];\mathbb{R}^d}}$, where $\dot{h}$ denotes the derivative of $h$.

For $x\in \mathbb{R}^m$, consider the stochastic differential equation (SDE):
\begin{equation}\label{origin}
X_t = x + \int_0^t b(X_s) \, \dd s + \int_0^t \sigma(X_s) \, \dd W_s, \quad t\in[0,1],
\end{equation}
where $\sigma=(\sigma_{ij})_{1\leq i\leq m,1\leq j\leq d} \colon \mathbb{R}^m \to \mathbb{R}^m \otimes \mathbb{R}^d$ and $b \colon \mathbb{R}^m \to \mathbb{R}^m$ are measurable functions. $b$, $\sigma$ and the derivative of $\sigma$, denoted by $\nabla\sigma$, are locally Lipschitz and satisfy some Lyapunov conditions; the precise assumptions on $b$ and $\sigma$ will be introduced in Section~\ref{section_3}. In the sequel, we denote the solution of~\eqref{origin} by $X\coloneqq\parens[\big]{X_t, t\in[0,1]}$.

The aim of this paper is to obtain the Stroock--Varadhan type support theorem for the SDE~\eqref{origin}, that is, we characterize the support of $\mathbb{P} \circ X^{-1}$ as the closure of the set $\braces[\big]{S(h);h\in\mathcal{H}}$ in $\mathcal{C}\parens[\big]{[0,1];\mathbb{R}^m}$, denoted by $\bar{\mathcal{S}}$, where $S(h)$ is the solution of the following deterministic equation:
\begin{equation}\label{appro2}
S(h)_t=x+ \int_0^t \bracks[\Bigg]{b\parens[\big]{S(h)_s}-\frac{1}{2}(\nabla\sigma)\sigma\parens[\big]{S(h)_s}} \, \dd s +\int_0^t \sigma\parens[\big]{S(h)_s}\dot{h}_s \, \dd s, \quad h\in \mathcal{H},
\end{equation}
where $(\nabla\sigma)\sigma$ can be expressed as $\parens[\big]{\sum_{l=1}^{d}\nabla \sigma_{kl} \sigma e_l}_{k=1,\dotsc,m}$, in which $\{e_1,\dotsc,e_d\}$ denotes the canonical basis of $\mathbb{R}^d$ and $\nabla\sigma_{kl} \colon \mathbb{R}^m \to \mathbb{R}^m$ is the vertical derivative of the $(k,l)$-coordinate of $\sigma$ for every $k \in \{1,\dotsc,m\}$ and $l \in \{1,\dotsc,d\}$. The support theorem is important in connection with the investigation of the accessibility, irreducibility and ergodicity of the Markov process generated by the solutions of SDE~\eqref{origin}.

The characterization of the topological support of the solutions was initially introduced by Stroock and Varadhan~\cite{DS} for SDEs with bounded and globally Lipschitz coefficients. GyÃ¶ngy and PrÃ¶hle~\cite{IT} later extended Stroock and Varadhan's result to SDEs with globally Lipschitz coefficients. Ben Arous, Gradinaru, and Ledoux~\cite{GM, GMM} obtained the support theorems for SDEs in a $\alpha$-HÃ¶lder space, requiring the coefficients $b$ and $\sigma$ to satisfy more strict conditions. Similar results as that in~\cite{DS} are found in~\cite{SSD}. We particularly want to mention the reference~\cite{AM} in which A.~Millet and M.~Sanz-SolÃ© proposed a simple, effective approach to obtain the support theorems for SDEs by proving Wong--Zakai type approximations for some associated SDEs.

In the work on support theorems so far, for technical reasons people always assume that the coefficients of the SDEs are globally Lipschitz. These restrictions exclude many important models, like Threshold Ornstein--Uhlenbeck process, stochastic SIR models, stochastic Lotka--Volterra systems, stochastic Duffing--van der Pol oscillator models, where the coefficients are of polynomial growth. The purpose of this paper is to extend the Stroock--Varadhan support theorem to SDEs under Lyapunov conditions. Our conditions are mild to allow coefficients of the SDEs to be locally Lipschitz with polynomial growth. In particular, the results can be applied to the interesting models mentioned above.

To obtain our main results, we adopt the same approach as that in~\cite{AM}. The crucial step is to obtain a Wong--Zakai type approximation for SDEs with local Lipschitz coefficients satisfying certain Lyapunov conditions. The idea is to introduce some localization arguments in order to utilize the existing results in the case where the coefficients of the SDEs are bounded and globally Lipschitz.

Before ending the introduction, let us briefly mention some relevant work on Wong--Zakai approximation of SDEs. Wong--Zakai approximation was introduced by Wong and Zakai in their pioneer work~\cite{WZ,WZ2} based on the idea that the noise in SDEs can be approximated by piecewise linear approximations of Brownian motion. Since then, there are a number of papers devoted to this topic. We mention the relevant work~\cite{CK,GX,KA,K,M,AM,NY,RW,SI,DS} and references therein. However, in all these works, the coefficients of the SDEs to required to be globally Lipschitz and bounded.

We would like to point out that apart from the application of Wong--Zakai approximations to prove the Stroock--Varadhan support theorems, it can be employed to derive some numerical approximation schemes for SDEs, which find many applications in electrical engineering, energy engineering, and related fields, see~\cite{SZM,WEG,MN}.

The paper is organized as follows. In Section~\ref{section_2}, we introduce the precise assumptions on $B$, $H$, $F$ and $G$ and prove the Wong--Zakai approximation results. In Section~\ref{section_3}, we prove the support theorem for SDEs. Some new applications are presented in Section~\ref{section_4}.

\section{Wong--Zakai approximations}\label{section_2}

In this section we will establish the Wong--Zakai approximations for SDEs with locally Lipschitz coefficients. Given a positive integer $n$, for $t \in[0,1]$, $\frac{k}{2^n}\leq t < \frac{k+1}{2^n}$, set
\begin{equation}
\twave_n = \frac{k}{2^n}, \qquad \underline{t}_n=\frac{k-1}{2^n}\vee 0,
\end{equation}
and define
\begin{equation}\label{wn}
W_t^n = W_{\underline{t}_n}+ 2^n(t-\twave_n)\bracks[\big]{W_{\twavesub_n} -W_{\underline{t}_n}}.
\end{equation}

Let $B \colon \mathbb{R}^m \to \mathbb{R}^m $, $H$, $G$ and $F \colon \mathbb{R}^m \to \mathbb{R}^m \otimes \mathbb{R}^d$ be given measurable mappings.

Introduce the following conditions:

\begin{enonce}{Assumption} \label{assump1}
Assume $B$, $H$, $F$ are continuous maps on\/ $\mathbb{R}^m$ and $G$ is a $\mathcal{C}^1$-map on\/ $\mathbb{R}^m$ satisfying:
\begin{enumerate}\romanenumi
\item $B$, $H, F$, $G$ and $\nabla G$ are locally Lipschitz;
\item there exists a Lyapunov function $V \in \mathcal{C}^2(\mathbb{R}^m;\mathbb{R}_+)$ and $\theta>0$, $\eta>0$ such that:
\begin{enumerate}\alphenumii
\item $\lim_{\abs{x}\to + \infty}V(x)=+\infty$;
\item the following holds:
\[
\begin{split}
J_1(x)
& \coloneqq\ \angles[\big]{B(x),\nabla V(x)} \\
& \quad + \frac{\theta}{2}\Trace\braces[\bigg]{\parens[\big]{H(x)^\ast\nabla^2V(x)H(x)}+\parens[\big]{G(x)^\ast\nabla^2V(x)G(x)} + \parens[\big]{F(x)^\ast\nabla^2V(x)F(x)}} \\
& \quad + \frac{\abs[\big]{\parens[\big]{H(x)+G(x)+F(x)}^\ast\nabla V(x)}^2}{\eta V(x)}\\
& \leq\ C\parens[\big]{1+V(x)};
\end{split}
\]
%\item $\Trace\braces[\bigg]{\parens[\big]{H(x)^\ast\nabla^2V(x)H(x)}+\parens[\big]{G(x)^\ast\nabla^2V(x)G(x)}+\parens[\big]{F(x)^\ast\nabla^2V(x)F(x)}}\geq -M-CV(x)$;
\item $\Trace\braces[\bigg]{\mk\parens[\big]{H(x)^\ast\nabla^2V(x)H(x)} \! + \! \parens[\big]{G(x)^\ast\nabla^2V(x)G(x)} \! + \! \parens[\big]{F(x)^\ast\nabla^2V(x)F(x)}\mk} \Mk \geq \Mk - \! M-CV(x)$;
\item the following holds:
\[
\begin{split}
J_2(x)
& \coloneqq\ \angles[\bigg]{B(x)+\nabla G(x)\bracks[\bigg]{F(x)+\frac{1}{2}G(x)},\nabla V(x)} \\
& \quad + \frac{\theta}{2}\Trace\braces[\bigg]{\parens[\big]{H(x)^\ast\nabla^2V(x)H(x)}+\parens[\big]{\parens[\big]{F(x)+G(x)}^\ast\nabla^2V(x)\parens[\big]{F(x)+G(x)}}} \\
& \quad + \frac{\abs[\big]{\parens[\big]{H(x)+G(x)+F(x)}^\ast\nabla V(x)}^2}{\eta V(x)}\\
& \leq \ C\parens[\big]{1+V(x)};
\end{split}
\]
\item $\Trace\braces[\bigg]{\parens[\big]{H(x)^\ast\nabla^2V(x)H(x)} +\parens[\big]{\parens[\big]{F(x)+G(x)}^\ast\nabla^2V(x)\parens[\big]{F(x)+G(x)}}}\geq -M-CV(x)$.
\end{enumerate}
\end{enumerate}
Here $\nabla V$, $\nabla^2 V$ and $\nabla G$ stand for the gradient vector, Hessian matrix of the function $V$ and the derivative of the function $G$, respectively; $\cdot^\ast$ denotes the transpose of the matrix; $C, M>0$ are some fixed constants.
\end{enonce}

For $h\in \mathcal{H}$, consider SDEs:
\begin{align}
Y_t^n &= x + \int_0^tB(Y_s^n) \, \dd s + \int_0^t H(Y_S^n)\dot{h}_s \, \dd s + \int_0^t G(Y_s^n)\dot{W}_s^n \, \dd s + \int_0^t F(Y_s^n) \, \dd W_s, \label{Yn}\\
Z_t &= \ x + \int_0^t B(Z_s) \, \dd s + \int_0^t H(Z_s)\dot{h}_s \, \dd s \nonumber \\
&\quad\ + \int_0^t \nabla G(Z_s)\bracks[\Bigg]{F(Z_s)+\frac{1}{2}G(Z_s)} \, \dd s + \int_0^t \bracks[\big]{F(Z_s)+G(Z_s)} \, \dd W_s, \label{Z}
\end{align}
where $(\nabla G) F$ can be expressed as $\parens[\big]{\sum_{l=1}^{d}\nabla G_{kl} F e_l}_{k=1,\dotsc,m}$, where $\{e_1,\dotsc,e_d\}$ denotes the canonical basis of $\mathbb{R}^d$ and $\nabla G_{kl} \colon \mathbb{R}^m \to \mathbb{R}^m$ is the vertical derivative of the $(k,l)$-coordinate of $G$ for every $k \in \{1,\dotsc,m\}$ and $l \in \{1,\dotsc,d\}$. It is known (see e.g.~\cite[Propositions~2.1 and~2.2]{JH}) that under Assumption~\ref{assump1} the SDEs~\eqref{Yn} and~\eqref{Z} admit unique solutions. The following result is the Wong--Zakai approximation.
\begin{theo}\label{theo_2.1}
Let $Y^n$, $Z$ be solutions of SDEs~\eqref{Yn} and~\eqref{Z}, respectively. Suppose Assumption~\ref{assump1} is in place. Then, for $\delta>0$
\begin{equation}\label{3}
\lim_{n\to \infty}\mathbb{P}\parens[\big]{\abs{Y^n-Z}_{\infty}>\delta}=0.\\
\end{equation}
\end{theo}

\begin{proof}
Without loss of generality, we assume $\delta<1$. For $R\in\mathbb{N}$, set
\[
\mathbb{B}_R=\braces[\bigg]{f: f\in \mathcal{C}\parens[\big]{[0,1];\mathbb{R}^m}, \; \abs{f}_{\infty}\leq R}.
\]
Since $Z\in \mathcal{C}\parens[\big]{[0,1];\mathbb{R}^m}$ a.e., we have,
\begin{equation}\label{4}
1=\mathbb{P}\parens[\big]{Z(\omega)\in \textstyle\bigcup_{R\in\mathbb{N}}\mathbb{B}_R}= \lim_{R\to \infty}\mathbb{P}\parens[\big]{Z(\omega)\in \mathbb{B}_R}.\\
\end{equation}
Let $\Omega_R\coloneqq\braces[\big]{\omega \in \Omega: Z(\omega)\in \mathbb{B}_R}$. In view of~\eqref{4}, for any given $\epsilon>0$, we can choose $R\in \mathbb{N}$ sufficiently large so that
\begin{equation}\label{5}
\mathbb{P}(\Omega_R^c)=1-\mathbb{P}(\Omega_R)<\frac{\epsilon}{2}.\\
\end{equation}
Take a smooth truncation function $\theta_R \colon \mathbb{R}^m\to \mathbb{R}$ such that $\theta_R\in [0,1]$ and that
\begin{equation}
\theta_R =
\begin{cases}
1	& \text{if $\abs{x}\leq R+1$},		\\
0	& \text{if $\abs{x}\geq 2(R+1)$}.	\\
\end{cases}
\end{equation}
Set $B_R(x)=\theta_R(x) B(x)$, $H_R(x)=\theta_R(x) H(x)$, $G_R(x)=\theta_R(x) G(x)$, $F_R(x)=$ $\theta_R(x) F(x)$.

It's easy to see that:
\begin{enumerate}\romanenumi
\item $B_R$, $H_R$, $G_R$, $F_R$ and $\nabla G_R$ are globally Lipschitz and bounded,
\item $B_R$, $H_R$, $G_R$, $F_R$ and $\nabla G_R$ coincide with $B, H, G, F$ and $\nabla G$ on the closed ball $\bar{B}(0,R+1)\subset \mathbb{R}^m$.
\end{enumerate}

For $R>0$, $h\in \mathcal{H}$, introduce the following SDEs:
\begin{align}
Z_t^R = & \ x + \int_0^t B_R(Z_s^R) \, \dd s + \int_0^t H_R(Z_s^R)\dot{h}_s \, \dd s \nonumber \\
&+ \int_0^t \nabla G_R(Z_s^R)\bracks[\Bigg]{F(Z_s)+\frac{1}{2}G_R(Z_s^R)} \, \dd s + \int_0^t \bracks[\bigg]{F_R(Z_s^R)+G_R(Z_s^R)} \, \dd W_s. \label{6} \\
Y_t^{n,R} = & \ x + \int_0^tB_R(Y_s^{n,R}) \, \dd s + \int_0^t H_R(Y_S^{n,R})\dot{h}_s \, \dd s + \int_0^t G_R(Y_s^{n,R})\dot{W}_s^n \, \dd s + \int_0^t F_R(Y_s^{n,R}) \, \dd W_s. \label{7}
\end{align}
Then, since $B_R$, $H_R$, $G_R$, $F_R$ and $\nabla G_R$ are globally Lipschitz and bounded, according to~\cite[Theorem~3.5]{AM}, we have
\begin{equation}\label{9-1}
\lim_{n\to\infty}\mathbb{P}\parens[\big]{\abs{Z^R-Y^{n,R}}_{\infty}>\delta}=0.
\end{equation}
For $\omega \in \Omega_R$, we have
\[
\abs[\big]{Z(\omega,\cdot\,)}_{\infty}\leq R.
\]
This indicates that on $\Omega_R$, $Z$ satisfies the same equation~\eqref{6} as $Z^R$. The uniqueness of solutions to equation~\eqref{6} implies that $Z^R(\omega,t)=Z(\omega,t)$ on $[0,1]$ for $\omega \in \Omega_R$.

For $R>0$, in view of~\eqref{9-1}, there exists $N_{\epsilon,R}\in \mathbb{N}$ such that for all $n\geq N_{\epsilon,R}$,
\begin{equation}\label{P}
\mathbb{P}\parens[\big]{\abs{Z^R-Y^{n,R}}_{\infty}>\delta}<\frac{\epsilon}{2}.\\
\end{equation}
For $n\geq N_{\epsilon,R}$, set
\begin{equation}\label{Ond}
\Omega_{n,R}\coloneqq\braces[\bigg]{\omega\in\Omega:\abs[\big]{Z^R(\omega,\cdot\,)-Y^{n,R}(\omega,\cdot\,)}_{\infty}>\delta}.
\end{equation}
We claim that if $n\geq N_{\epsilon,R}$ and if $\omega \in (\Omega_R^c\cup \Omega_{n,R})^c=\Omega_R\cap\Omega_{n,R}^c$, then
\begin{equation}\label{10}
Y^{n,R}(\omega,t)=Y^n(\omega,t) \quad \text{on $[0,1]$}.
\end{equation}
Indeed, for $\omega \in \Omega_R$, we have $\abs[\big]{Z^R(\omega,\cdot\,)}_{\infty}=\abs[\big]{Z(\omega,\cdot\,)}_{\infty}\leq R$. If $\omega \in \Omega_{n,R}^c$, we have $\abs[\big]{Z^R(\omega,\cdot\,)-Y^{n,R}(\omega,\cdot\,)}_{\infty}\leq\delta<1$. Therefore, on the set $\Omega_R\cap\Omega_{n,R}^c$,
\begin{equation}\label{11}
\abs[\big]{Y^{n,R}(\omega,\cdot\,)}_{\infty}\leq \abs[\big]{Z^R(\omega,\cdot\,)-Y^{n,R}(\omega,\cdot\,)}_{\infty} + \abs[\big]{Z^R(\omega,\cdot\,)}_{\infty}\leq R+1.
\end{equation}
For $R>0$, define the stopping times:
\begin{align}
&\tau^{R+1}=\inf\braces[\big]{t, \abs{Y^n_t}\geq R+1},\\
&\tilde{\tau}^{R+1}=\inf\braces[\big]{t, \abs{Y^{n,R}_t}\geq R+1}.
\end{align}
Set $\tau=\tau^{R+1}\wedge \tilde{\tau}^{R+1}$. Then, $Y^n_{t\wedge \tau}$ and $Y^{n,R}_{t\wedge \tau}$ are solutions of the same equation~\eqref{7}. Hence by the uniqueness,
\begin{equation}\label{8}
Y^n(\omega,t\wedge \tau) = Y^{n,R}(\omega,t\wedge \tau) \quad \text{on $[0,1]$}.
\end{equation}
This further implies that $\tau=\tau^{R+1}= \tilde{\tau}^{R+1}$. By~\eqref{11}, we see that
\begin{equation}\label{8-1}
\Omega_R\cap\Omega_{n,\delta}^c \subset \braces[]{\tilde{\tau}^{R+1}>1}=\{\tau>1\}=\{\tau^{R+1}>1\}.
\end{equation}
Combining~\eqref{8} with~\eqref{8-1} yields that for $\omega \in (\Omega_R^c\cup \Omega_{n,\delta})^c=\Omega_R\cap\Omega_{n,\delta}^c$,
\begin{equation}\label{8-2}
Y^{n,R}(\omega,t)=Y^n(\omega,t) \quad \text{on $[0,1]$}.
\end{equation}
Combining~\eqref{Ond} and~\eqref{10} together, we deduce that for $n\geq N_{\epsilon,R}$, $\omega \in (\Omega_R^c\cup \Omega_{n,\delta})^c=\Omega_R\cap\Omega_{n,\delta}^c$,
\begin{equation}
\abs[\big]{Y^n(\omega,\cdot\,)-Z(\omega,\cdot\,)}_{\infty}=\abs[\big]{Y^{n,R}(\omega,\cdot\,)-Z^R(\omega,\cdot\,)}_{\infty}\leq \delta.
\end{equation}
Consequently, by~\eqref{5}, \eqref{P},
\[
\begin{split}
\mathbb{P}\parens[\big]{\abs{Y^n-Z}_{\infty}>\delta}
& \leq\ \mathbb{P}(\Omega_R^c\cup \Omega_{n,\delta})				\\
& \leq\ \mathbb{P}(\Omega_R^c) + \mathbb{P}(\Omega_{n,\delta})	\\
& \leq\ \frac{\epsilon}{2} + \frac{\epsilon}{2}					\\
& = \epsilon.
\end{split}
\]
Since $\epsilon$ is arbitrary, we conclude that
\[
\lim_{n\to\infty}\mathbb{P}\parens[\big]{\abs{Y^n-Z}_{\infty}>\delta}=0
\]
completing the proof.
\end{proof}

\section{Support theorems}\label{section_3}

Now we turn back to the stochastic differential equation~\eqref{origin} and present a support theorem for the solution. Regarding the coefficients $b$, $\sigma$, we introduce the following condition.

\begin{enonce}
{Assumption} \label{assump2}
Assume $b$ is a continuous map on $\mathbb{R}^m$ and $\sigma$ is a $\mathcal{C}^1$-map on $\mathbb{R}^m$ satisfying:
\begin{enumerate}\romanenumi
\item both $b$, $\sigma$ and $\nabla \sigma$ are locally Lipschitz;
\item there exists a Lyapunov function $V \in \mathcal{C}^2(\mathbb{R}^m;\mathbb{R}_+)$ and $\theta>0$, $\eta>0$ such that
\begin{align*}
& \lim_{\abs{x}\to + \infty}V(x)=+\infty, \\
& J_1(x)\coloneqq\angles[\big]{b(x),\nabla V(x)} + \frac{\theta}{2}\Trace\parens[\big]{\sigma^\ast(x)\nabla^2V(x)\sigma(x)} + \frac{\abs[\big]{\sigma^\ast(x)\nabla V(x)}^2}{\eta V(x)} \\
& \hspace{2.5em} \leq C\parens[\big]{1+V(x)}, \\
& J_2(x) \coloneqq \angles[\bigg]{b(x)-\frac{1}{2}\nabla\sigma\sigma(x),\nabla V(x)} + \frac{\theta}{2}\Trace\parens[\big]{\sigma^\ast(x)\nabla^2V(x)\sigma(x)} + \frac{\abs[\big]{\sigma^\ast(x)\nabla V(x)}^2}{\eta V(x)} \\
& \hspace{2.5em} \leq C\parens[\big]{1+V(x)}, \\
& \Trace\parens[\big]{\sigma^\ast(x)\nabla^2V(x)\sigma(x)}\geq -M-CV(x).
\end{align*}
\end{enumerate}
Here $\nabla V$, $\nabla^2 V$ and $\nabla \sigma$ stand for the gradient vector, Hessian matrix of the function $V$ and the derivative of the function $\sigma$, respectively; $\sigma^\ast(x)$ denotes the transpose of $\sigma(x)$; $C,M>0$ are some fixed constants.
\end{enonce}
\begin{rema}
\mbox{}
\begin{enumerate}
\item The assumption on $J_2(x)$ is proposed to ensure the existence of a unique solution to equation~\eqref{appro2}.
\item The Lyapunov conditions mentioned above are not actually used in our proof. Indeed, once the equation~\eqref{origin} possesses a unique solution and the truncated coefficients are globally Lipschitz and bounded, the Wong--Zakai and support results can be established in a similar way.
\end{enumerate}
\end{rema}

Recall the stochastic differential equation:
\begin{equation}\label{3-1}
X_t = x + \int_0^t b(X_s) \, \dd s + \int_0^t \sigma(X_s) \, \dd Ws, \quad t\in[0,1].
\end{equation}
The existence and uniqueness of the solution $X$ under the Assumption~\ref{assump2} follows from~\cite{JH}.

The aim of this section is to characterize the support of $\mathbb{P} \circ X^{-1}$ as the closure $\mathcal{S}$ of the set $\braces[\big]{S(h);h\in\mathcal{H}}$ in $\mathcal{C}\parens[\big]{[0,1];\mathbb{R}^m}$.

First we recall the following proposition from~\cite{AM}.
\begin{prop}\label{prop_3.1}
Consider a measurable map $V \colon \Omega\to E$, where $\parens[\big]{E,\abs{\,\cdot\,}_E}$ is a separable Banach space.
\begin{enumerate}
\item Let $\zeta_1 \colon \mathcal{H}\to E$ be a measurable map, and let $H_n \colon \Omega\to \mathcal{H}$ be a sequence of random variables such that for any $\epsilon>0$,
\begin{equation}
\lim_{n\to \infty}\mathbb{P}\parens[\big]{\abs[\big]{V(\omega)-\zeta_1\parens[\big]{H_n(\omega)}}_E>\epsilon}=0.
\end{equation}
Then
\begin{equation}
\support(\mathbb{P}\circ V^{-1})\subset \bar{\zeta_1(\mathcal{H})}.
\end{equation}
\item Let $\zeta_2 \colon \mathcal{H}\to E$ be a map, and for fixed $h$ let $T_n^h \colon \Omega \to \Omega$ be a sequence of measurable transformations such that $\mathbb{P}\circ(T_n^h)^{-1}\ll \mathbb{P}$, and for any $\epsilon >0$,
\begin{equation}
\limsup_{n\to \infty} \mathbb{P}\parens[\big]{\abs[\big]{V\parens[\big]{T_n^h(\omega)}-\zeta_2(h)}_E<\epsilon}>0.
\end{equation}
Then\/ $\support(\mathbb{P}\circ V^{-1})\supset \bar{\zeta_2(\mathcal{H})}$.
\end{enumerate}
\end{prop}

The following result is the support theorem.
\begin{theo}\label{support}
Suppose Assumption~\ref{assump2} holds. Let $X$ and $S(h)$ be the solutions to equations~(\ref{3-1}) and~(\ref{appro2}). Then\/ $\supp(\mathbb{P}\circ X^{-1})=\bar{\mathcal{S}}$, where $\bar{\mathcal{S}}$ denotes the closure of $\mathcal{S}=\braces[\big]{S(h);h\in\mathcal{H}}$ in the space $\mathcal{C}\parens[\big]{[0,1];\mathbb{R}^m}$ and $\supp(\mathbb{P}\circ X^{-1})$ denotes the support of the distribution $\mathbb{P}\circ X^{-1}$.
\end{theo}
\begin{proof}
We will apply Proposition~\ref{prop_3.1}. To this end, we take $E=\mathcal{C}\parens[\big]{[0,1];\mathbb{R}^m}$, $V=X$, $\zeta_1=\zeta_2=S(\,\cdot\,)$, $H_n(\omega)= \omega^n$ and $T_n^h(\omega)=\omega-\omega^n+h$, where $\omega^n$ is defined as~\eqref{wn}. Then Girsanov's theorem implies that $\mathbb{P}\circ(T_n^h)^{-1}\ll \mathbb{P}$. Thus, according to Proposition~\ref{prop_3.1}, the equality $\supp(\mathbb{P}\circ X^{-1})=\bar{\mathcal{S}}$ will follow from the following approximation results, for every $\delta>0$:
\begin{align}
&\lim_{n\to \infty}\mathbb{P}\parens[\big]{\abs[\big]{X(\omega)-S(\omega^n)}_{\infty}>\delta}=0,			\label{1}\\
&\lim_{n\to \infty} \mathbb{P}\parens[\big]{\abs[\big]{X(\omega-\omega^n+h)-S(h)}_{\infty}>\delta}>0.		\label{2}
\end{align}
Where $S(\omega^n)$ is the solution of equation
\begin{equation}\label{001}
S(\omega^n)_t= x +\int_0^t \bracks[\Bigg]{b\parens[\big]{S(\omega^n)_s}-\frac{1}{2}(\nabla\sigma)\sigma\parens[\big]{S(\omega^n)_s}} \, \dd s +\int_0^t \sigma\parens[\big]{S(\omega^n)_s}\dot{W}^n_s \, \dd s.
\end{equation}
On the other hand, approximations of stochastic integrals by Riemann sums imply that $X^n(\omega)\coloneqq X(\omega-\omega^n+h)$ is the solution of the following stochastic differential equation:
\begin{equation}\label{002}
X^n_t=x +\int_0^t b(X_s^n) \, \dd s + \int_0^t \sigma(X_s^n)\dot{h}_s \, \dd s - \int_0^t \sigma(X_s^n)\dot{W}_s^n \, \dd s + \int_0^t \sigma(X_s^n) \, \dd W_s.
\end{equation}
By a close examination, we find that both $X^n$ and $ S(\omega^n)$ are particular cases of solutions of the stochastic differential equation~\eqref{Yn} in Section~\ref{section_2}. Actually, setting $B=b-\frac{1}{2}(\nabla\sigma)\sigma$, $H=0$, $G=\sigma$ and $F=0$ in the equation~\eqref{3} we obtain~\eqref{1}; while setting $B=b$, $H=\sigma$, $G=-\sigma$ and $F=\sigma$ gives~\eqref{2}. Therefore, \eqref{1} and~\eqref{2} are the particular cases of the convergence stated in Theorem~\ref{theo_2.1}:
\begin{equation}
\lim_{n\to \infty}\mathbb{P}\parens[\big]{\abs{Y^n-Z}_{\infty}>\delta}=0.
\end{equation}
The proof is complete.
\end{proof}

\vspace{-.75\baselineskip}
\section{Examples}\label{section_4}
\vspace{-.25\baselineskip}

The Assumption~\ref{assump2} is very mild to include many interesting models. In this section we provide some examples to which the main results apply.

\begin{exem}
Consider the following one-dimensional SDE:
\begin{equation}
\dd x_t = -x^3_t \, \dd t + x^2_t  \, \dd B_t.
\end{equation}
In this case, $b(x)=-x^3$, $\sigma(x)=x^2$ and $\nabla\sigma(x)=2x$. If we take $\theta=1$, $\eta=4$ and $V(x)=x^2$, then
\begin{align*}
& \lim_{\abs{x}\to + \infty}V(x)=+\infty,\\
& J(x)\coloneqq\angles[\big]{b(x),\nabla V(x)} + \frac{\theta}{2}\Trace\parens[\big]{\sigma^\ast(x)\nabla^2V(x)\sigma(x)} + \frac{\abs[\big]{\sigma^\ast(x)\nabla V(x)}^2}{\eta V(x)}=0 \leq 1+V(x),\\
& \angles[\bigg]{-\frac{1}{2}\nabla\sigma\sigma,\nabla V(x)} \leq0,\\
& \Trace\parens[\big]{\sigma^\ast(x)\nabla^2V(x)\sigma(x)}=2x^4 \geq -V(x).
\end{align*}
\end{exem}
This shows that Assumption~\ref{assump2} holds. We can now apply Theorem~\ref{support} to get the following result.
\begin{prop}
Let $\mathcal{H}$ denote the Cameron--Martin space. For $h\in \mathcal{H}$, let $S(h)$ be the solution of the differential equation:
\[
S(h)_t=x_0+ \int_0^t -2S(h)_s^3  \, \dd s +\int_0^t S(h)_s^2 \dot{h}_s \, \dd s.
\]
Then\/ $\supp(\mathbb{P}\circ x^{-1})=\bar{\mathcal{S}}$, where $\bar{\mathcal{S}}$ denotes the closure of $\mathcal{S}=\braces[\big]{S(h);h\in\mathcal{H}}$ in $\mathcal{C}\parens[\big]{[0,1];\mathbb{R}}$.
\end{prop}

The next three examples are taken from~\cite{JH}.
\begin{exem}[Stochastic Duffing--van der Pol oscillator model]
The Duffing--van der Pol oscillator equation unifies both the Duffing equation and the van der Pol equation describing a self-oscillating triode/diode circuit. The stochastic version of the model is given by the following SDE (see~\cite{SMA}).
\begin{align*}
& \ddot{X}^{x,1}_t = \alpha_2\dot{X}^{x,1}_t - \alpha_1 X^{x,1}_t-\alpha_3(X^{x,1}_t)^2\dot{X}^{x,1}_t - (X^{x,1}_t)^3 + g(X^{x,1}_t)\dot{W}_t,\\
& X^{x,1}_0 = x_1, \quad \dot{X}^{x,1}_0=x_2,
\end{align*}
where $\alpha_1,\alpha_2,\alpha_3\in (0,\infty).$ Here we assume $\abs[\big]{g(x)}^2\leq\eta_0+\eta_1\abs{x}^4$, $\eta_0,\eta_1>0$ and both $g(x)$ and $g'(x)$ are locally Lipschitz. Setting $X^{x,2}_t\coloneqq\dot{X}^{x,1}_t$, then the above equation is equivalent to following system of SDEs:
\begin{align*}
& \dd X^{x,1}_t 	= X^{x,2}_t \, \dd t,\\
& \dd X^{x,2}_t	= \bracks[\big]{\alpha_2 X^{x,2}_t - \alpha_1 X^{x,1}_t-\alpha_3(X^{x,1}_t)^2 X^{x,2}_t - (X^{x,1}_t)^3} +g(X^{x,1}_t) \, \dd W_t,\\
& X^{x,1}_0 =x_1, \quad X^{x,2} = x_2.
\end{align*}
For $x=(x_1,x_2)\in\mathbb{R}^2$, set $b(x)=\parens[\big]{x_2,\alpha_2x_2-\alpha_1x_1-\alpha_3(x_1)^2x_2-(x_1)^3}^T$ and $\sigma(x)=\parens[\big]{0,g(x_1)}^T$. Define $V(x)=\frac{(x_1)^4}{2}+\alpha_1(x_1)^2 +(x_2)^2$, $\theta=\eta=1$. Then:
\begin{align*}
& \lim_{\abs{x}\to + \infty}V(x)=+\infty,\\
& J(x) \coloneqq \angles[\big]{b(x),\nabla V(x)} + \frac{1}{2}\Trace\parens[\big]{\sigma^\ast(x)\nabla^2V(x)\sigma(x)} + \frac{\abs[\big]{\sigma^\ast(x)\nabla V(x)}^2}{V(x)}\\
& \hspace{2.2em} = x_2\parens[\big]{2(x_1)^3 +2\alpha_1x_1} + 2x_2\parens[\big]{\alpha_2 x_2-\alpha_1x_1-\alpha_3(x_1)^2x_2-(x_1)^3}\\
& \hspace{2.2em} \quad + \abs[\big]{g(x_1)}^2 +\frac{4(x_2)^2\abs[\big]{g(x_1)}^2}{\frac{(x_1)^4}{2}+\alpha_1(x_1)^2 +(x_2)^2}\\
& \hspace{2.2em} = 2\alpha_2(x_2)^2-2\alpha_3(x_1)^2(x_2)^2 + \abs[\big]{g(x_1)}^2 + \frac{4(x_2)^2\abs[\big]{g(x_1)}^2}{\frac{(x_1)^4}{2}+\alpha_1(x_1)^2 +(x_2)^2}\\
& \hspace{2.2em} \leq \eta_0 + 2\alpha_2(x_2)^2 + \eta_1(x_1)^4 + \frac{4\eta_0(x_2)^2 + \eta_1(x_1)^4(x_2)^2}{\frac{(x_1)^4}{2}+\alpha_1(x_1)^2 +(x_2)^2}\\
& \hspace{2.2em} \leq 5\eta_0 + (8\eta_1 +2\alpha_2)(x_2)^2+\eta_1(x_1)^4\\
& \hspace{2.2em} \leq (5\eta_0+10\eta_1+2\alpha_2)\parens[\big]{1+V(x)},\\
& \angles[\bigg]{-\frac{1}{2}\nabla\sigma\sigma,\nabla V(x)} =0, \\
& \Trace\parens[\big]{\sigma^\ast(x)\nabla^2V(x)\sigma(x)}=2\abs[\big]{g(x_1)}^2 \geq -V(x).
\end{align*}
Hence Assumption~\ref{assump2} holds and we have the following result.
\begin{prop}
Let $\mathcal{H}$ denote the Cameron--Martin space. For $h\in \mathcal{H}$, let $S(h)=\parens[\big]{S(h)_1,S(h)_2}^T$ be the solution of the differential equation:
\begin{align*}
S(h)_{1,t}	& = x_1+ \int_0^t S(h)_{2,s}  \, \dd s,\\
S(h)_{1,t}	& = x_2+ \int_0^t \alpha_2 S(h)_{2,s}- \alpha_1 S(h)_{1,s}-\alpha_3\parens[\big]{s(h)_{1,s}}^2S(h)_{2,s} - \parens[\big]{S(h)_{1,s}}^3  \, \dd s +\int_0^t g\parens[\big]{S(h)_{1,s}}\dot{h}_s \, \dd s.
\end{align*}
Then\/ $\supp(\mathbb{P}\circ X^{-1})=\bar{\mathcal{S}}$, where $\bar{\mathcal{S}}$ denotes the closure of $\mathcal{S}=\braces[\big]{S(h);h\in\mathcal{H}}$ in $\mathcal{C}\parens[\big]{[0,1];\mathbb{R}^2}$.
\end{prop}
\end{exem}

\begin{exem}[Stochastic Lotka--Volterra (LV) systems]
The Lotka--Volterra systems play an important role in game theory, population dynamics etc.\ (see~\cite{JK}). Here we consider the three-dimensional Stratonovich stochastic competitive LV system:
\begin{align*}
\dd y_1 &= y_1(r-a_{11}y_1-a_{12}y_2-a_{13}y_3) \, \dd t + \gamma y_1\circ  \, \dd B_t,\\
\dd y_2 &= y_2(r-a_{21}y_1-a_{22}y_2-a_{23}y_3) \, \dd t + \gamma y_2\circ  \, \dd B_t,\\
\dd y_3 &= y_3(r-a_{31}y_1-a_{32}y_2-a_{33}y_3) \, \dd t + \gamma y_3\circ  \, \dd B_t,
\end{align*}
where $\sigma,r>0$, $a_{ij}>0$, $i,j=1,2,3$; $\gamma$ is parameter and initial data $\parens[\big]{y_1(0),y_2(0),y_3(0)}\in(0,+\infty)^3$. According to~\cite[Theorem~3.2]{LZ}, we know $y(t)=\parens[\big]{y_1(t),y_2(t),y_3(t)}\in(0,+\infty)^3$ for all $t>0$. And the above system is equivalent to the ItÃ´ stochastic Lotka--Volterra system:
\[
\dd y_i = y_i\parens[\Bigg]{r + \frac{\gamma^2}{2} -\sum_{j=1}^3a_{ij}y_j} \, \dd t + \gamma y_i \, \dd B_t,\quad i=1,2,3.
\]
Set $b(y)=\parens[\big]{y_1\parens[\big]{r + \frac{\gamma^2}{2} -\sum_{j=1}^3a_{1j}y_j},y_2\parens[\big]{r + \frac{\gamma^2}{2} -\sum_{j=1}^3a_{2j}y_j},y_3\parens[\big]{r + \frac{\gamma^2}{2} -\sum_{j=1}^3a_{3j}y_j}}^T$, $\sigma(y)=(\gamma y_1,\gamma y_2,\gamma y_3)^T$. Let $V(y)=\abs{y}^2$. Then:
\begin{align*}
& \lim_{\abs{y}\to + \infty}V(y)=+\infty,\\
& J(y)\coloneqq\angles[\big]{b(y),\nabla V(y)} + \frac{\theta}{2}\Trace\parens[\big]{\sigma^\ast(y)\nabla^2V(y)\sigma(y)} + \frac{\abs[\big]{\sigma^\ast(y)\nabla V(y)}^2}{\eta V(y)}\\
& \hspace{2.2em} =\sum_{i=1}^3 2y_i^2\parens[\Bigg]{r + \frac{\gamma^2}{2} -\sum_{j=1}^3a_{ij}y_j}+\theta\gamma^2V(y) +\frac{4\gamma^2}{\eta}V(y)\\
& \hspace{2.2em} \leq 2V(y)\parens[\Bigg]{r + \frac{\gamma^2}{2}} + \theta\gamma^2V(y) +\frac{4\gamma^2}{\eta}V(y)\\
& \hspace{2.2em}\leq C\parens[\big]{1+V(y)},\\
& \angles[\bigg]{-\frac{1}{2}\nabla\sigma\sigma,\nabla V(x)} \leq0,\\
& \Trace\parens[\big]{\sigma^\ast(y)\nabla^2V(y)\sigma(y)}=2\gamma^2V(y)\geq 0.
\end{align*}
Hence, Assumption~\ref{assump2} holds. Apply Theorem~\ref{support} to get the following.
\begin{prop}
Let $\mathcal{H}$ denote the Cameron--Martin space. For $h\in \mathcal{H}$, let
\[
S(h)=\parens[\big]{s(h)_1,S(h)_2,S(h)_3}^T
\]
be the solution of the differential equation:
\begin{align*}
S(h)_{1,t}	& = y_1(0)+ \int_0^t S(h)_{1,s}\parens[\Bigg]{r -\sum_{j=1}^3a_{1j}S(h)_{j,s}} \, \dd s + \int_0^t\gamma S(h)_{1,s}\dot{h}_s \, \dd s, \\
S(h)_{2,t}	& = y_2(0)+ \int_0^t S(h)_{2,s}\parens[\Bigg]{r -\sum_{j=1}^3a_{2j}S(h)_{j,s}} \, \dd s + \int_0^t\gamma S(h)_{2,s}\dot{h}_s \, \dd s, \\
S(h)_{3,t}	& = y_2(0)+ \int_0^t S(h)_{3,s}\parens[\Bigg]{r -\sum_{j=1}^3a_{3j}S(h)_{j,s}} \, \dd s + \int_0^t\gamma S(h)_{3,s}\dot{h}_s \, \dd s.
\end{align*}
Then\/ $\supp(\mathbb{P}\circ X^{-1})=\bar{\mathcal{S}}$, where $\bar{\mathcal{S}}$ denotes the closure of $\mathcal{S}=\braces[\big]{S(h);h\in\mathcal{H}}$ in $\mathcal{C}\parens[\big]{[0,1];\mathbb{R}^3}$.
\end{prop}
\end{exem}

\begin{exem}[Stochastic SIR model]
The SIR model from epidemiology for the total number of susceptible, infected and recovered individuals has been introduced by Anderson and May~\cite{RM2}.

Here we consider the following stochastic SIR model:
\begin{align*}
& \dd X_t^{x,1} = \parens[\big]{-\alpha X_t^{x,1}X_t^{x,2}-\kappa X_t^{x,1} +\kappa} \, \dd t -\beta X_t^{x,1}X_t^{x,2} \, \dd W_t,\\
& \dd X_t^{x,2} = \parens[\big]{\alpha X_t^{x,1}X_t^{x,2}-(\gamma +\kappa) X_t^{x,2}} \, \dd t +\beta X_t^{x,1}X_t^{x,2} \, \dd W_t,\\
& \dd X_t^{x,3} = (\gamma X_t^{x,2}-\kappa X_t^{x,3}) \, \dd t,\\
& X_0^{x,1} = x_1,\quad X_0^{x,2} = x_2,\quad X_0^{x,3} = x_3,
\end{align*}
where $\alpha,\beta,\gamma,\kappa\in (0,\infty)$ and $x=(x_1,x_2,x_3)\in[0,\infty)^3$.

For $x=(x_1,x_2,x_3)\in[0,\infty)^3$, set $b(x)=\parens[\big]{-\alpha x_1x_2-\kappa x_1 +\kappa, \alpha x_1x_2-(\gamma + \kappa)x_2,\gamma x_2-\kappa x_3}^T$ and $\sigma(x)=(-\beta x_1x_2,\beta x_1x_2,0)^T.$ It's easy to see that $b$ and $\sigma$ are local Lipschitz continuous and satisfy the Assumption~\ref{assump2} with $V(x)=(x_1 + x_2 - 1)^2$ for any positive $\theta$ and $\eta$. Then applying Theorem~\ref{support}, we have:
\begin{prop}
Let $\mathcal{H}$ denote the Cameron--Martin space. For $h\in \mathcal{H}$, let
\[
S(h)=\parens[\big]{s(h)_1,S(h)_2,S(h)_3}^T
\]
be the solution of the differential equation:
\begin{align*}
S(h)_{1,t}	& = x_1+ \int_0^t \parens[\Bigg]{-\alpha S(h)_{1,s}S(h)_{2,s}-\frac{\beta^2}{2}\parens[\big]{S(h)_{1,s}S(h)_{2,s}}\parens[\big]{S(h)_{2,s}-S(h)_{1,s}}-\kappa S(h)_{1,s} +\kappa} \, \dd s\\
			& \quad - \int_0^t \beta S(h)_{1,s}S(h)_{2,s}\dot{h}_s \, \dd s,\\
S(h)_{2,t}	& = x_2+ \int_0^t \parens[\Bigg]{\alpha S(h)_{1,s}S(h)_{2,s}+\frac{\beta^2}{2}\parens[\big]{S(h)_{1,s}S(h)_{2,s}}\parens[\big]{S(h)_{2,s}-S(h)_{1,s}}-(\gamma +\kappa) S(h)_{2,s}} \, \dd s\\
			& \quad+ \int_0^t\beta S(h)_{1,s}S(h)_{2,s}\dot{h}_s \, \dd s,\\
S(h)_{3,t}	& = x_3+ \int_0^t \parens[\big]{\gamma S(h)_{2,s}-\kappa S(h)_{3,s}} \, \dd s.
\end{align*}
Then\/ $\supp(\mathbb{P}\circ X^{-1})=\bar{\mathcal{S}}$, where $\bar{\mathcal{S}}$ denotes the closure of $\mathcal{S}=\braces[\big]{S(h);h\in\mathcal{H}}$ in $\mathcal{C}\parens[\big]{[0,1];\mathbb{R}^3}$.
\end{prop}
\end{exem}

\begin{exem}[Threshold Ornstein--Uhlenbeck processes]
\begin{equation}\label{exp3.4}
 \dd X_t=\sum_{i=1}^{n}(\beta_i -\alpha_i X_t)I_{\braces[]{\theta_{i-1}\leq X_t <\theta_i}} \, \dd t + \sigma \, \dd B_t,\\
\end{equation}
where $ \beta_1,\dotsc,\beta_n,\alpha_1,\dotsc,\alpha_n,-\infty=\theta_0<\theta_1<\dotsb<\theta_{n-1}<\theta_n=\infty$ are constants. Set $b(x)=\sum_{i=1}^{n}(\beta_i -\alpha_ix)I_{\braces[]{\theta_{i-1}\leq x < \theta_i}}$ and $\sigma(x)=\sigma$. It's clear that~\eqref{exp3.4} satisfy the Assumption~\ref{assump2} with $V(x)=\abs{x}^2.$ Then applying Theorem~\ref{support}, we have:
\begin{prop}
Let $\mathcal{H}$ denote the Cameron--Martin space. For $h\in \mathcal{H}$, let $S(h)$ be the solution of the differential equation:
\[
S(h)_t = X_0+ \int_0^t \sum_{i=1}^{n}\parens[\big]{\beta_i -\alpha_i S(h)_s}I_{\braces[]{\theta_{i-1}\leq S(h)_s < \theta_i}} \, \dd s + \int_0^t\sigma\dot{h}_s \, \dd s.
\]
Then\/ $\supp(\mathbb{P}\circ X^{-1})=\bar{\mathcal{S}}$, where $\bar{\mathcal{S}}$ denotes the closure of $\mathcal{S}=\braces[\big]{S(h);h\in\mathcal{H}}$ in $\mathcal{C}\parens[\big]{[0,1];\mathbb{R}}$.
\end{prop}
\end{exem}

\section*{Declaration of interests}

The authors do not work for, advise, own shares in, or receive funds from any organization that could benefit from this article, and have declared no affiliations other than their research organizations.

\printbibliography

\end{document}